Author's response to reviews

Title: Using a social marketing framework to evaluate recruitment of a prospective study of genetic counseling and testing for the deaf community

Authors:

Yoko Kobayashi (deafyoko@gmail.com)
Patrick Boudreault (pboudreault@usfca.edu)
Karin Hill (karinh@gmail.com)
Janet S Sinsheimer (janet@mednet.ucla.edu)
Christina GS Palmer (cpalmer@mednet.ucla.edu)

Version: 2 Date: 11 April 2013

Author's response to reviews:

Dear Editor,

Per your request, we have revised our manuscript to more explicitly describe in the Methods section the ethical approvals we received for the conduct of our study. Specifically, the original sentence has been revised as follows: "This research study was conducted in compliance with the Helsinki Declaration and approved by the institutional review board at the University of California Los Angeles (#10-001193) and by the Committee for Protection of Human Subjects at the California State University-Northridge."

Please do not hesitate to contact me if you any questions.

Best,
Christina Palmer, PhD
UCLA